ARTICLE

Real-World Evidence of Tofacinitib in Ulcerative Colitis: Short-Term and Long-Term Effectiveness and Safety

María Chaparro, MD, PhD^{1,*}, Diana Acosta, Sc^{1,*}, Cristina Rodríguez, MD², Francisco Mesonero, MD³, Miren Vicuña, MD², Manuel Barreiro-de Acosta, MD, PhD⁴, Agnès Fernández-Clotet, MD⁵, Álvaro Hernández Martínez, MD, PhD⁶, Maite Arroyo, MD, PhD⁷, Isabel Vera, MD, PhD⁸, Alexandra Ruiz-Cerulla, MD⁹, Beatriz Sicilia, MD, PhD¹⁰, M. José Cabello Tapia, MD, PhD¹¹, Carmen Muñoz Villafranca, MD¹², Jesús Castro-Poceiro, MD¹³, Jesús Martínez Cadilla, MD¹⁴, Mónica Sierra-Ausín, MD¹⁵, Juan María Vázquez Morón, MD, PhD¹⁶, Raquel Vicente Lidón, MD¹⁷, Fernando Bermejo, MD, PhD¹⁸, Vanesa Royo, MD¹⁹, Margalida Calafat, MD, PhD²⁰, Carlos González-Muñoza, MD²¹, Eduardo Leo Carnerero, MD²², Noemi Manceñido Marcos, MD, PhD²³, Leyanira Torrealba, MD²⁴, Horacio Alonso-Galán, MD²⁵, José Manuel Benítez, MD²⁶, Yolanda Ber Nieto, MD²⁷, M. Teresa Diz-Lois Palomares, MD²⁸, María José García, MD²⁹, José Fernando Muñoz, MD³⁰, Edisa María Armesto González, MD³¹, Xavier Calvet, MD, PhD³², Alejandro Hernández-Camba, MD, PhD³³, Rosa Eva Madrigal Domínguez, MD³⁴, Luis Menchén, MD, PhD³⁵, José Lázaro Pérez Calle, MD³⁶, Marta Piqueras, MD³⁷, Carmen Dueñas Sadornil, MD³⁸, Belén Botella, MD³⁹, Teresa de Jesús Martínez-Pérez, MD⁴⁰, Laura Ramos, MD⁴¹, María Carmen Rodríguez-Grau, MD, PhD⁴², Elena San Miguel, MD⁴³, José Luis Fernández Forcelledo, MD⁴⁴, Paola María Fradejas Salazar, MD⁴⁵, Marifé García-Sepulcre, MD, PhD⁴⁶, Ana Gutiérrez, MD, PhD⁴⁷, Jordina Llaó, MD, PhD⁴⁸, Eva Sesé Abizanda, MD, PhD⁴⁹, Maia Boscá-Watts, MD, PhD⁵⁰, Eduardo Iyo, MD⁵¹, Alma Keco-Huerga, MD⁵², Carmen Martínez Bonil, MD⁵³, Elena Peña González, MD, PhD⁵⁴, Pablo Pérez-Galindo, MD⁵⁵, Pilar Varela, MD⁵⁶ and Javier P. Gisbert, MD, PhD¹ on Behalf of To-ReWard Study Group

INTRODUCTION: The objective of this study was to assess the durability, short-term and long-term effectiveness, and safety of tofacitinib in ulcerative colitis (UC) in clinical practice.

METHODS: This is a retrospective multicenter study including patients with UC who had received the first tofacitinib dose at least 8 weeks before the inclusion. Clinical effectiveness was based on partial Mayo score.

¹Hospital Universitario de La Princesa, Instituto de Investigación Sanitaria Princesa (IIS-Princesa), Universidad Autónoma de Madrid (UAM), Centro de Investigación Biomédica en Red de Enfermedades Hepáticas y Digestivas (CIBEREHD), Madrid, Spain; ²Hospital Universitario de Navarra (HUN), Instituto de Investigación Sanitaria de Navarra (IdiSNA), Pamplona, Spain; ³Hospital Universitario Ramón y Cajal, Madrid, Spain; ⁴Hospital Clínico Universitario de Santiago de Compostela Santiago de Compostela, Spain; ⁵Hospital Clinic of Barcelona, Institut d'Investigacions Biomèdiques August Pi i Sunyer (IDIBAPS) and CIBERehd, Barcelona, Spain; ⁶Hospital Universitario Torrecárdenas, Almería, Spain; ⁷Hospital Clínico Universitario Lozano Blesa and Fundación del Instituto de Investigación Sanitaria de Aragón (IIS Aragón) and CIBEREHD, Zaragoza, Spain; ⁸Hospital Universitario Puerta de Hierro Majadahonda, Madrid, Spain; ⁹Hospital Universitario de Bellvitge, L'Hospitalet de Llobregat, Barcelona, Spain; ¹⁰Hospital Universitario de Burgos, Burgos, Spain; ¹¹Hospital Universitario Virgen de las Nieves, Granada, Spain;¹²Hospital Universitario de Basurto, Bilbao, Spain;¹³Hospital Sant Joan Despí-Moisès Broggi, Barcelona, Spain;¹⁴Hospital Álvaro Cunqueiro, Xerencia Xestión Integrada de Vigo, SERGAS. Grupo de Investigación en Patología Digestiva, Instituto de Investigación Sanitaria Galicia Sur (IIS Galicia Sur), SERGAS-UVIGO, Vigo, Spain; ¹⁵Complejo Asistencial Universitario de León, Spain; ¹⁶Hospital Universitario Juan Ramón Jiménez, Huelva, Spain; ¹⁷Hospital Universitario Miguel Servet, Zaragoza, Spain; ¹⁸Hospital Universitario de Fuenlabrada and IdiPAZ, Madrid, Spain; ¹⁹Hospital Universitari Son Espases, Palma, Islas Baleares, Spain; ²⁰Hospital Universitari Germans Trias i Pujol and CIBEREHD, Badalona, Spain; ²¹Hospital de la Santa Creu i Sant Pau, Barcelona, Spain; ²²Hospital Universitario Virgen del Rocío, Sevilla, Spain; ²³Hospital Universitario Infanta Sofía, Madrid, Spain; ²⁴Hospital Universitario Dr. Josep Trueta, Girona, Spain; ²⁶Hospital Universitario Donostia, Instituto Biodonostia, San Sebastián, Guipúzcoa, Spain; ²⁶Hospital Universitario Reina Sofía, IMIBIC, Córdoba, Spain; ²⁷Hospital San Jorge, Huesca, Spain; ²⁸Complejo Hospitalario Universitario A Coruña, A Coruña, Spain; ²⁹Hospital Universitario Marqués de Valdecilla and Instituto de Investigación Sanitaria Valdecilla (IDÍVAL), Santander, Spain; ³⁰Hospital Universitario de Salamanca, Salamanca, Spain; ³¹Hospital San Agustín, Avilés, Asturias, Spain: ³²Hospital Universitari Parc Taulí, Sabadell, Universitat Autònoma de Barcelona and CIBERehd, Spain; ³³Complejo Hospitalario Universitario Nuestra Señora de Candelaria, Santa Cruz de Tenerife, Spain; ³⁴Hospital Clínico Universitario de Valladolid, Valladolid, Spain; ³⁵Hospital General Universitario e Instituto de Investigación Sanitaria Gregorio Marañón, y Universidad Complutense, Madrid, Spain; ³⁶Hospital Universitario Fundación Alcorcón, Madrid, Spain; ³⁷Consorci Sanitari de Terrassa, Barcelona, Spain; ³⁸Hospital San Pedro de Alcántara, Cáceres, Spain; ³⁹Hospital Universitario Infanta Cristina, Madrid, Spain; ⁴⁰Hospital Virgen de la Luz, Cuenca, Spain; ⁴¹Hospital Universitario de Canarias, Santa Cruz de Tenerife, Spain; ⁴²Hospital Universitario del Henares, Madrid, Spain; ⁴³Hospital Universitario de Getafe, Madrid, Spain; ⁴⁴Hospital Sierrallana de Torrelavega, Cantabria, Spain; ⁴⁵Hospital Virgen de la Concha, Zamora, Spain; ⁴⁶Hospital General Universitario de Elche, Alicante, Spain; ⁴⁷Hospital General Universitario de Alicante, Instituto de Investigación Sanitaria y Biomédica de Alicante (ISABIAL), and CIBERehd, Alicante, Spain; ⁴⁸Althaia Xarxa Assistencial Universitaria de Manresa, Barcelona, Spain; ⁴⁹Hospital Universitario Arnau de Vilanova, Lleida, Spain; ⁵⁰Hospital Clínico de Valencia, Valencia, Spain; ⁵¹Hospital Comarcal de Inca, Islas Baleares, Spain; ⁵²Hospital de Valme, Sevilla, Spain; ⁵³Hospital General Universitario Los Arcos del Mar Menor, Murcia, Spain; ⁵⁴Hospital Royo Villanova, Zaragoza, Spain; ⁵⁵Complexo Hospitalario Universitario de Pontevedra, Galicia, Spain; ⁵⁶Hospital Universitario de Cabueñes, Gijón, Spain. Correspondence: María Chaparro, MD, PhD. E-mail: mariachs2005@gmail.com. *María Chaparro, Diana Acosta contributed equally to this work.

Received July 13, 2022; accepted October 12, 2022; published online January 30, 2023

© 2023 by The American College of Gastroenterology

The American Journal of GASTROENTEROLOGY

RESULTS: A total of 408 patients were included. Of them, 184 (45%) withdrew tofacitinib during follow-up (mean = 18 months). The probability of maintaining tofacitinib was 67% at 6 m, 58% at 12 m, and 49% at 24 m. The main reason for tofacitinib withdrawal was primary nonresponse (44%). Older age at the start of tofacitinib and a higher severity of clinical activity were associated with tofacitinib withdrawal. The proportion of patients in remission was 38% at week 4, 45% at week 8, and 47% at week 16. Having moderate-to-severe vs mild disease activity at baseline and older age at tofacitinib start were associated with a lower and higher likelihood of remission at week 8, respectively. Of 171 patients in remission at week 8, 83 (49%) relapsed. The probability of maintaining response was 66% at 6 m and 54% at 12 m. There were 93 adverse events related to tofacitinib treatment (including 2 pulmonary thromboembolisms [in patients with risk factors] and 2 peripheral vascular thrombosis), and 29 led to tofacitinib discontinuation.

DISCUSSION: Tofacitinib is effective in both short-term and long-term in patients with UC. The safety profile is similar to that previously reported.

KEY WORDS: ulcerative colitis; tofacitinib; real-world evidence

SUPPLEMENTARY MATERIAL accompanies this paper at http://links.lww.com/AJG/C840

Am J Gastroenterol 2023;00:1-11. https://doi.org/10.14309/ajg.00000000002145

INTRODUCTION

Ulcerative colitis (UC) is a chronic inflammatory bowel disease with a relapsing-remitting pattern that causes bloody diarrhea, leading to organ damage and impaired quality of life. The primary goals of therapy in UC are reducing the mucosal inflammation and maintaining symptom remission, though these aims are not achieved in all patients (1,2).

In the last 2 decades, the introduction of targeted and biological therapies has changed the natural history of UC. Nevertheless, up to 30% of patients do not respond to a first-line treatment (primary nonresponders), and approximately 20% per year lose response after an initial improvement (secondary nonresponders), thereby requiring a dose escalation or a switch to another drug class (3–10). In addition to the initially described tumor necrosis factor (TNF), there are several other cytokine pathways involved in the development of UC, which have led to the development of target-specific drugs.

Tofacitinib is an oral synthetic small-molecule Janus kinase inhibitor. Janus kinase are downstream signaling molecules of many cytokine pathways involved in inflammatory bowel disease (11). Tofacitinib was approved for the treatment of moderate-tosevere UC based on the efficacy and safety data shown in the UC clinical development program. The results of phase 3 clinical trials have confirmed the superiority of tofacitinib over placebo in induction treatment and maintenance of clinical remission in patients with moderate-to-severe UC (12).

The use of drugs in clinical trials differs from that of the routine clinical practice in several aspects such as patient characteristics (patients are frequently more refractory to treatments and with more comorbidities in real-life practice), thus limiting the generalization of clinical trial results. Noninterventional studies, on the contrary, provide complementary information to clinical trials on the effectiveness and the way of use of treatments in real clinical practice, both for effectiveness and safety, is still quite limited (13). Studies published so far include a limited number of patients with a limited follow-up (below 1 year), which makes it difficult to get robust conclusions and to examine certain aspects of the use of this drug.

Our objective was to understand in depth the role of tofacitinib in the treatment of UC in clinical practice and to provide useful data to know its real benefit. To reach our goal, we aimed to describe the durability of tofacitinib treatment and factors conditioning drug interruption in clinical practice, the short-term and long-term effectiveness, the dose adjustments, and the safety profile in a real-life, large, multicenter nationwide study. We anticipate that our results will provide useful data for the management of patients with UC in real life and will help to position tofacitinib in the therapeutic algorithm of UC.

METHODS

Study design

This is a retrospective, multicenter, noninterventional study to assess the retention of tofacitinib treatment in patients with UC. Every patient who met the criteria in each participating center was included. Patients were followed up until the last tofacitinib dose or the last visit, whichever came first. Data were remotely monitored to assess data quality. Because this is an observational study, the schedule of tofacitinib administration was not predefined and was decided by the clinician responsible for patients' treatment. The project was approved by the Ethics Committee of Hospital Universitario de La Princesa.

Patient population

The study population consisted of adult patients who had received at least 1 dose of tofacitinib due to UC at least 8 weeks before the start date of recruitment. Patients treated with tofacitinib for any indication other than UC, with previous colectomy, and/or those who had been involved in a clinical trial with tofacitinib were excluded.

Data collection

Patient demographic and clinical characteristics were collected: sex, age, smoking, age at diagnosis, disease extent, extraintestinal manifestations (EIM) and immune-mediated inflammatory diseases (IMID), previous surgery for UC, concomitant use of steroids and immunomodulators at the beginning of and during follow-up, previous treatments for UC, reasons for discontinuation of previous treatments for UC (immunomodulators and biologic agents), start date of tofacitinib therapy, response to tofacitinib, clinical activity at baseline and during follow-up, concomitant medication for UC, date of discontinuation (when it occurred), reason for discontinuation (lack of primary response, relapse, patient choice, adverse event, surgery for UC worsening, others, unknown reason), dosing regimen during maintenance, treatment after loss of response to tofacitinib (if any), dose increase (if any), dose decrease (if any), response after dose optimization, evolution of EIM and IMID, surgery for UC, hospitalizations (due to UC or for other reasons), and adverse events. In addition, information about endoscopic activity (endoscopic subscore of the Mayo index) or biological markers (such as C-reactive protein [CRP] or fecal calprotectin concentration) was requested from clinicians responsible for patients' treatment, when available.

Data were collected and managed using REDCap electronic data capture tools hosted at (www.aegastro.es). Asociación Española de Gastroenterología is a nonprofit Scientific and Medical Society focused on Gastroenterology, which provides this service free of charge, with the sole aim of promoting independent investigator-driven research.

Tofacitinib durability and effectiveness measurements

The durability of tofacitinib treatment was calculated considering the entire period under tofacitinib treatment: from the first dose to the last dose with this drug. Patients included in this study had to have received tofacitinib for clinical indication. Some patients might have received the treatment being in remission due to intolerance to or contraindication of other drugs. Only patients with active disease (partial Mayo score [PMS] > 2) were considered in the short-term effectiveness analysis.

Active disease was defined as a PMS >2. When endoscopy was available, the severity was graded by local investigators as quiescent, mild, moderate, or severe. The severity of clinical activity was rated based on the PMS.

Clinical remission was defined as a PMS ≤ 2 . Clinical response was defined as a reduction in the PMS ≥ 3 points and a decrease of at least 30% from baseline, with a decrease of ≥ 1 point on the rectal bleeding subscale (absolute score 0–1). Relapse was defined as a worsening in the patient's symptoms leading the doctor to intensify the treatment dose, add another medication for the control of UC, or switch to another treatment or surgery.

Evolution of EIM and IMID

The evolution of previous EIM and IMID was evaluated based on clinicians' judgment. The onset of new EIM and IMID under tofacitinib treatment was recorded.

Safety assessments

All adverse events occurring during tofacitinib treatment were registered and their relationship with tofacitinib administration, based on clinicians' criteria, were recorded. Serious adverse events were defined according to the International Council for Harmonisation of Technical Requirements for Pharmaceuticals for Human Use and European Union guidelines on pharmacovigilance of medicinal products for human use. Major cardiovascular adverse events were defined as death from cardiovascular causes, nonfatal myocardial infarction, or nonfatal stroke.

Statistical analysis

For categorical variables, percentages were calculated (with their 95% confidence intervals [CI]). The descriptive analysis of quantitative variables calculated the mean and SD, or the median and interquartile range (IQR), depending on whether the variables were normally distributed or not. In the univariate analysis, categorical variables were compared using the χ^2 test and quantitative variables using the appropriate test. For short-term effectiveness evaluation, only patients with PMS >2 at baseline were considered. Short-term effectiveness was evaluated at weeks 4, 8, and 16. Variables associated with the likelihood of treatment response after induction were identified using a logistic regression model. In patients who discontinued tofacitinib owing to lack of therapeutic effect, an adverse event, or worsening of UC before their last visit, the last observation carried forward method was used to impute missing values at subsequent time points for both short-term and long-term evaluation.

The Kaplan-Meier method, where patients who discontinued to facitinib for any reason were censored during discontinuation, was used to evaluate the long-term durability of to facitinib treatment (main outcome). In addition, we analyzed the cumulative incidence of relapse among patients who reached remission at week 8. Any differences between survival curves were evaluated with the log-rank test. Stepwise multivariate analysis using the Cox model was performed to identify factors associated with to facitinib discontinuation or relapse over time. In the log-rank test and multivariate analysis, statistical significance was considered when P < 0.05. The cumulative incidence of adverse events was calculated; in addition, for the most relevant adverse events, the incidence rates were calculated and expressed as patients with first events per 100 patient-years.

RESULTS

A total of 408 patients were included. The main characteristics of the study population are summarized in Table 1. A total of 234 patients (57%) had extensive colitis, and 151 (37%) had left-sided colitis. Almost all patients, but 26 (6.4%), had clinically active disease defined as a PMS >2 at tofacitinib start; 69 (17%) had mild activity and 313 (77%) moderate-to-severe activity. The mean age was 44 years, the median time from UC diagnosis was 7.5 years, 45% had comorbidities (see Supplementary Table 1, http://links.lww.com/AJG/C840), and 30% had cardiovascular and thrombotic risk factors (see Supplementary Table 2, http://links.lww.com/AJG/C840). Regarding previous treatment for UC, 97% had been exposed to biologics and more than 70% to 2 or more biologics before tofacitinib treatment (see Supplementary Table 3, http://links.lww.com/AJG/C840).

Durability of tofacitinib treatment in patients with UC

A total of 184 patients (45%) withdrew tofacitinib treatment during follow-up (median = 18 months, IQR = 10–29 months). The median time of exposure to tofacitinib was 8.5 months (IQR = 4–18 months). The incidence rate of tofacitinib discontinuation was 41% per patient-year of follow-up (95% CI = 36%–48%). The survival curve of tofacitinib treatment is shown in Figure 1. The probability of maintaining tofacitinib treatment was 91% at 1 month, 67% at 6 months, 58% at 12 months, 53% at 18 months, 49% at 24 months, 46% at 30 months, and 45% at 36 months. Reasons for tofacitinib withdrawal were as follows: 81 (44%) primary nonresponse, 48 (26%) loss of response, 29 (16%) adverse events, 19 (10%) partial response, and 7 (3.8%) medical decision (Figure 2). After tofacitinib withdrawal, 74 patients

© 2023 by The American College of Gastroenterology

The American Journal of GASTROENTEROLOGY

Table 1. Characteristics of the study population	n
Variables	N = 408
Age (yr), mean (SD)	44 (15)
Median time of follow-up (mo) (IQR)	18 (10–2
Median time from UC diagnosis (yr) (IQR)	7.5 (4–13
Male sex, n (%)	232 (57)
Comorbidities, n (%)	182 (45)
Cardiovascular and thrombotic risk factors, n (%)	124 (30)
Family history of IBD, n (%)	61 (15)
Ulcerative colitis extension	
Extensive colitis, n (%)	234 (57)
Left-sided colitis, n (%)	151 (37)
Proctitis, n (%)	23 (6)
Extraintestinal manifestations, n (%)	115 (28)
Previous biologic agents, n (%)	397 (97)
Anti-TNF, n (%)	377 (92)
Vedolizumab, n (%)	298 (73)
Ustekinumab, n (%)	24 (5.9)
Mean no. of previous biologic agents (SD)	2 (0.9)
Median partial mayo score at baseline (IQR)	6 (5–7)
Severe endoscopic activity, n (%)	140 (55)
Anemia at baseline, n (%)	132 (32)
Concomitant mesalazine, n (%)	163 (40)
IBD, inflammatory bowel disease; IQR, interquartile range; T factor.	TNF, tumor necrosis

(40%) were treated with ustekinumab, 36 (20%) underwent surgery, 29 (16%) received vedolizumab, 21 (11%) an anti-TNF agent, and 24 (13%) other therapeutic options.

In the univariate analysis, the mean age of patients who withdrew the treatment was significantly lower than that of those who maintained tofacitinib (41 vs 47 years, P < 0.05). The severity of clinical activity of UC at baseline was associated with a lower probability of tofacitinib survival: stool frequency (P <0.001), bloody stools (P < 0.001), and severity based on the PMS



Figure 1. Survival curve of tofacitinib treatment in ulcerative colitis.



Figure 2. Timing and reasons for tofacitinib withdrawal. AA, adverse events; LR, loss of response; MD, medical decision; PNR, primary nonresponse; PR, partial response.

(P < 0.001). Other variables such as sex, smoking habits, thromboembolic risk factors, family history of the disease, UC extension, EIM, type of previous treatments for UC, number and type of previous biologics, anemia at baseline, severity of endoscopic activity, or concomitant treatment with mesalamine were not associated with tofacitinib survival.

In the multivariate analysis, older age at the start of tofacitinib was associated with a lower likelihood of tofacitinib withdrawal

The American Journal of GASTROENTEROLOGY

Copyright © 2023 by The American College of Gastroenterology. Unauthorized reproduction of this article is prohibited.

5) 0–29)



Figure 3. Short-term effectiveness of tofacitinib in ulcerative colitis.

(hazard ratio [HR] = 0.98, 95% CI = 0.97–0.99), while the severity of clinical activity at baseline based on the PMS was associated with a higher likelihood of tofacitinib withdrawal (mild vs remission: HR = 1.5, 95% CI = 0.5–4, and moderate-to-severe vs remission: HR = 3, 95% CI = 1.2–7.4).

Short-term effectiveness

A total of 382 patients had clinical activity of the disease at baseline (PMS >2) and were included in the effectiveness analysis. Short-term effectiveness is shown in Figure 3; of note, 38% of patients were in remission at week 4, 45% at week 8, and 47% at



Figure 4. Proportion of patients with ulcerative colitis in remission under tofacitinib treatment.

© 2023 by The American College of Gastroenterology

The American Journal of GASTROENTEROLOGY



Figure 5. Survival curve of maintenance of tofacitinib effectiveness in patients with ulcerative colitis.

week 16. Of the patients with no response at week 4, 18% reached remission at week 8, and 26% at week 16. Of the patients with nonresponse at week 8, 19% reached remission at week 16.

Week 4

In the univariate analysis, the severity of clinical activity at baseline was the only variable associated with remission at week 4, and this finding was confirmed in the multivariate analysis, where the severity of the disease at baseline (moderate-to-severe vs mild) was the only variable associated with the likelihood of achieving remission at week 4 (odds ratio [OR] = 0.2, 95% CI = 0.1–0.4).

Week 8

In the univariate analysis, only the age at the start of tofacitinib treatment and the severity of clinical activity at baseline were associated with remission at week 8. In the multivariate analysis, having moderate-to-severe vs mild disease activity at baseline (OR = 0.2, 95% CI = 0.1-0.4) and older age at tofacitinib start (OR = 1.01, 95% CI = 1.002-1.03) were the only variables associated with the likelihood of achieving remission at week 8.

Week 16

In the univariate analysis, the presence of EIM, the age at the start of tofacitinib treatment, and the severity of clinical activity at baseline were associated with remission at week 16. In the multivariate analysis, having moderate-to-severe vs mild disease activity at baseline (OR = 0.4, 95% CI = 0.2-0.6) was the only variable associated with the likelihood of achieving remission at week 16.

Remission during follow-up

The proportion of patients in remission at each time point is represented in Figure 4. The proportion of patients in remission was above 40% during follow-up.

Relapse

Patients with active disease (PMS >2) at baseline who achieved remission at week 8 were included in the relapsing analysis. A total of 171 patients met these criteria; of them, 83 (48.5%) relapsed. The probability of maintaining remission was 90% at 3 months, 73% at 6 months, 57% at 12 months, 46% at 18 months, and 41% at 21 months (Figure 5). The incidence rate of relapse was 56% (95% CI = 44%–70%) per patient-year of follow-up. Neither in the univariate nor in the multivariate analysis, factors associated with relapse could be identified. In fact, the cumulative incidence of relapse was similar in patients who deescalated the dose after the induction than in those who maintained the initial dose. Tofacitinib dose was escalated in 55 (66%) of those patients who relapsed; 33 (60%) reached remission, 12 (22%) had response, and 10 (18%) did not respond.

Dose adjustments

A total of 261 patients (64%) had at least 1 dose change during tofacitinib treatment. The incidence rate of relapse was similar in patients with active disease at baseline who started with 10 mg bid dose, reached remission at week 8, and maintained the same dose over time than in those who were changed to the maintenance dose (5 mg bid). A total of 144 patients (35%) had 1 change, 80 (20%) 2 changes, 31 (7.6%) 3 changes, 4 (1%) 4 changes, and 2 (0.5%) 5 changes. Tofacitinib doses over time are summarized in

The American Journal of GASTROENTEROLOGY

VOLUME 00 | MONTH 2023 www.amjgastro.com

Table 2. Extraintestinal manifestations and immune-mediated inflammatory diseases in the study population

Previous nonactive at tofacitinib start	n (%)	Worsening attrib trea	utable to tofacitinib	
Juvenile idiopathic arthritis	1 (0.2)	None		
Peripheral arthropathy	15 (3.7)	3 (20%) worsened		
Rheumatoid arthritis	1 (0.2)	None		
Primary sclerosing cholangitis	4 (1)	Ν	lone	
Autoimmune thyroid disease	2 (0.5)	Ν	lone	
Episcleritis	4(1)	1 (25%)) worsened	
Erythema nodosum	12 (2.9)	Ν	lone	
Scleritis	1 (0.2)	Ν	lone	
Multiple sclerosis	1 (0.2)	Ν	lone	
Axial spondiloarthropathy	11 (2.7)	Ν	lone	
Aphthous stomatitis	3 (0.7)	Ν	lone	
Systemic lupus erythematosus	1 (0.2)	Ν	lone	
Pyoderma gangrenosum	2 (0.5)	Ν	lone	
Psoriasis	13 (3.2)	Ν	lone	
Uveitis	1 (0.2)	Ν	lone	
Vasculitis	1 (0.2)	Ν	lone	
Others	5 (1)	Ν	lone	
Previous active at tofacitinib start	n (%)	Outcome attributable	to tofacitinib treatment	
Juvenile idiopathic arthritis	1 (0.2)	No d	change	
Peripheral arthropathy	32 (7.8)	8 (25%) remission, 1 11 (34.4	3 (40.6%) improvement,) no change	
Rheumatoid arthritis	4 (1)	1 (25%) remission, 1 (25%) improvement, 1 (25%) no change, 1 (25%) worsening		
Primary sclerosing cholangitis	1 (0.2)	No d	change	
Autoimmune thyroid disease	1 (0.2)	No change		
Erythema nodosum	2 (0.5)	2 (100%) remission		
Axial spondiloarthropathy	15 (3)	3 (23%) remission, 7 (54%) improvement, 5 (33%) no change, 1 (7%) worsen		
Systemic lupus erythematosus	1 (0.2)	No	change	
Pyoderma gangrenosum	1 (0.2)	1 (100%) improvement		
Psoriasis	1 (0.2)	No d	change	
Uveitis	1 (0.2)	No e	change	
Others	2 (0.5)	No d	change	
New onset under tofacitinib treatment	N (%)	Attitude toward tofacitinib	Outcome	
Peripheral arthropathy	4 (1)	No change	2 no change 2 remission	
Erythema nodosum	1 (0.2)	No change	1 remission	
Aphthous stomatitis	1 (0.2)	No change	1 remission	
Vasculitis	1 (0.2)	Withdrawal	1 improvement	
IgA pemphigus	1 (0.2)	No change	1 no change	

Supplementary Digital Content (see Supplementary Table 4, http://links.lww.com/AJG/C840); of note, the proportion of patients on 10 mg bid was more than 40% in all time points during follow-up.

EIM and IMID

In our study, approximately one-third of patients had EIM and/or IMID (active or not) at the start of tofacitinib treatment (prevalence and outcomes of previous nonactive EIM or IMID,

The American Journal of GASTROENTEROLOGY

DISEASE
BOWEL
TORY
MMA
INFLA

Table 3. Adverse events under tofacitinib treatment

Adverse event	n (% of the overall cohort)	Severe	Serious adverse event	Attributed to tofacitinib	Tofacitinib discontinuation	Resolved without sequelae
Anemia, n (%)	15 (3.7)	1 (6.7)	1 (6.7)	6 (40)	1(6.7)	10 (67)
Nonmelanoma skin cancer, n (%)	2 (0.5)	1 (50)	1 (50)	0 (0)	0 (0)	2 (100)
Cardiovascular events, n (%)	3 (0.5)	3 (100)	3 (100)	0 (0)	1 (50)	2 (66)
Herpes zoster, n (%)	11 (2.6)	0 (0)	0 (0)	7 (64)	1 (9)	9 (82)
Herpes (other than herpes zoster), n (%)	10 (2.5)	0 (0)	0 (0)	8 (80)	0 (0)	10 (100)
Hypercholesterolemia, n (%)	27 (6.6)	0 (0)	0 (0)	24 (89)	0 (0)	14 (52)
Hypertriglyceridemia, n (%)	5 (1.2)	1 (20)	0 (0)	3 (60)	0 (0)	4 (80)
Infections, n (%)	46 (11)	7 (15)	18 (39)	18 (39)	9 (20)	40 (87)
Lymphopenia, n (%)	7 (1.7)	0 (0)	0 (0)	6 (86)	1 (14)	3 (43)
Neoplasias, n (%)	1 (0.2)	1 (100)	1 (100)	0 (0)	0 (0)	0 (0)
Pulmonary thromboembolisms, n (%)	2 (0.5)	1 (50)	2 (100)	2 (100)	2 (100)	1 (50)
Peripheral venous thromboembolisms, n (%)	2 (0.5)	0 (0)	0 (0)	2 (100)	1 (33)	2 (100)
Others, n (%)	59 (15)	8 (14)	9 (15)	27 (46)	13 (22)	38 (64)

3 exitus (heart stroke, complications from COVID-19, breast cancer) considered nonrelated to tofacitinib by the clinicians responsible for patients' care.

prevalence and outcomes of active EIM and IMD, and attitude toward tofacitinib treatment and outcomes are summarized in Table 2). The most frequent EIM at tofacitinib start was

Table 4. Incidence rates of the most relevant adverse events during tofacitinib treatment

Type of adverse event	N	Incidence rate (95% confidence interval) (cases per 100 patient-yr)	
Major cardiovascular adverse events All Attributable to tofacitinib	1 0	0.23 (0.03–1.66) —	
Cardiac adverse events All Attributable to tofacitinib	3 0	0.70 (0.22–2.18) —	
Herpes zoster All Attributable to tofacitinib	11 7	2.64 (1.46–4.77) 1.67 (0.79–3.50)	
Pulmonary thromboembolism All ^a Attributable to tofacitinib	2 2	0.46 (0.11–1.87) 0.46 (0.11–1.87)	
Peripheral venous thrombosis All ^a Attributable to tofacitinib	2 2	0.46 (0.11–1.87) 0.46 (0.11–1.87)	
Serious infections All Attributable to tofacitinib	18 7	4.35 (2.74–6.91) 1.67 (0.79–3.5)	
Nonmelanoma skin cancer All Attributable to tofacitinib	2 0	0.47 (0.11–1.87)	
^a All of them had been considered attributable to tofacitinib.			

peripheral arthropathy. A total of 15 patients had nonactive peripheral arthropathy during tofacitinib initiation, and 3 of them had a worsening that was attributed to treatment. One patient with a history of episcleritis also worsened, but all other EIM inactive at baseline remained inactive. Among the active EIM tofacitinib initiation, the most frequent one was peripheral arthropathy (32 patients): of them, 8 remitted and 13 improved with tofacitinib. A total of 15 patients had active axial arthropathy, of which 10 patients achieved remission of symptoms or improved. The occurrence of *de novo* EIM or IMID was rare and led to withdrawal of tofacitinib in only 1 case (vasculitis), with resolution of symptoms after treatment discontinuation.

Hospitalizations

A total of 91 patients (22%) were hospitalized during the study period; 63 patients (69%) due to UC.

Safety

A total of 152 patients (37%) had at least 1 adverse event during follow-up. The severity, the relation to tofacitinib, the attitude toward the drug, and the outcomes are summarized in Table 3, while the incidence rate of the main adverse events is summarized in Table 4.

There were 35 patients (8%) who had serious adverse events under tofacitinib treatment, infections and cardiovascular events being the most frequent ones (Table 3). The major cardiovascular event and thromboembolic events are summarized in Supplementary Digital Content (see Supplementary text, http://links. lww.com/AJG/C840).

A total of 46 patients (11%) had infections in our cohort: 11 gastrointestinal infections, 10 SARS-CoV-2 infections, 7 respiratory infections, 4 urinary infections, 2 skin infections, 2 perianal abscesses, 1 bartolinitis, 1 sepsis of unknown origin, and 8 patients had other infections. Of those infections, there were 18 that met the criteria of serious adverse events (39% of all the infections): 7 gastrointestinal infections, 7 SARS-CoV-2

The American Journal of GASTROENTEROLOGY

VOLUME 00 | MONTH 2023 www.amjgastro.com

infections, 2 perianal abscesses, 1 orchiepididymitis, and 1 sepsis of unknown origin.

DISCUSSION

Tofacitinib has been shown to be effective in the treatment of UC in randomized clinical trials (11,12,14,15). In this clinical practice study, which includes the largest number with the longest follow-up ever published, we have been able to demonstrate that tofacitinib is also effective in real life even in very refractory patients.

In our cohort, the incidence rate of tofacitinib discontinuation was approximately 40% per patient-year of follow-up (which is slightly higher to that described for anti-TNF agents). This may reflect a more refractory cohort but may also be a consequence of the availability of a number of targeted drugs that allow for more ambitious treatment goals in these patients. In fact, after discontinuation of tofacitinib, only 20% of patients underwent surgery; the remaining 80% received other medical treatments. Of note, most treatment discontinuations occurred within the first 6 months of treatment, and the main cause of treatment withdrawal was primary nonresponse. However, after the first 6 months of treatment, the ability to maintain tofacitinib treatment remained stable over time. Therefore, according to the results of our study, patients who respond and maintain treatment beyond 6 months have a high probability that the benefit of tofacitinib will be maintained in the long-term.

The persistence of tofacitinib treatment in the long-term was recently evaluated in the Oral Clinical Trials for tofAcitinib in ulceratiVE colitis (OCTAVE) Open trial (15). Authors observed that, overall, 79% of patients dropped out from the trial during a total of 2,440 patient-years of exposure (approximately 30% per patient-year). These figures were lower in patients who responded to tofacitinib treatment, where the overall persistence rate was 73.9% and 54.5% at 2 and 5 years, respectively (15). Regarding clinical practice, Lucaciu et al (13) performed a meta-analysis on the real-world evidence of tofacitinib treatment up to 2021; treatment discontinuation was reported in 35% of patients across 8 studies (the median follow-up was 31 weeks).

In general, the main reason for tofacitinib discontinuation was lack of/insufficient primary response both in randomized clinical trials and in clinical practice setting (13–26). Regarding predictive factors of tofacitinib discontinuation, in the OCTAVE open trial, younger age (younger than 40 years), female sex, tofacitinib 10 mg bid at baseline, and previous anti-TNF failure were associated with an increased risk of discontinuation in patients who responded to tofacitinib (15). Our results confirm that, in clinical practice, older age was associated with a higher probability of maintaining the treatment, while the severity of clinical activity at baseline was associated with a higher likelihood of tofacitinib withdrawal.

We also evaluated tofacitinib short-term effectiveness. We observed that almost 40% of patients at week 4 and almost 50% at weeks 8 and 16 reached remission. At week 8, 18.5% of patients in OCTAVE 1 and 16.6% of patients in OCTAVE 2 were in clinical remission (remission was assessed with full Mayo score with centrally assessed endoscopic subscores) (12). Other clinical practice studies have shown similar results to ours. In the meta-analysis by Lucaciu et al (13), the percentage of patients in clinical remission after induction (weeks 8, 12, or 14) was 37% (range: 31%–51%), although the heterogeneity of the included studies was significant.

Regarding predictive factors of short-term effectiveness, we observed that the severity of the disease at baseline impaired the likelihood of achieving remission, while older age was associated with better results. These findings agreed with those described by Honap et al (21) who observed that younger age at treatment initiation and elevated CRP at baseline were independently associated with primary nonresponse. These findings are in line with those previously described in clinical trials, where both CRP and PMS at baseline were associated with short-term response, confirming that patients with higher inflammatory burden are at a higher risk of treatment failure (27). On the contrary, in OCTAVE 1 and 2 trials, the treatment effect was similar between those who had received previous treatment with an anti-TNF agent and those who had not (12). The impact of previous biologic failure cannot be assessed in clinical practice studies where more than 90% of the patients had been exposed to biologics previously to the initiation of tofacitinib.

The durability of response to treatment in the long-term is of upmost importance. In clinical trials, remission at 52 weeks occurred in 34.3% of the patients in the 5 mg bid group and in 40.6% in the 10 mg bid group (12). To our knowledge, our study is the first one providing long-term data in real life; we observed that more than 40% of patients were in remission in the long-term (44% at 12 months). Of note, approximately 50% of our patients were receiving 10 mg bid in the long run.

We found that approximately 50% of patients who achieved remission at week 8 relapsed during follow-up. In two-thirds of patients, the dose was escalated from 5 to 10 mg bid, and 60% of them regained remission. The response to dose escalation was greater in our study than in the OCTAVE study, where after dose escalation, 35% and 49% were in remission, at months 2 and 12, respectively (28). Our results suggest that dose escalation after relapsing is a feasible and effective strategy to recapture remission in clinical practice.

The comparison between clinical trials and clinical practice must be undertaken with caution. The design and modality of a real-world observational study are clearly different from those of a randomized clinical trial. Indeed, the outcomes of the OCTAVE trials were based on full Mayo score with centrally assessed endoscopic subscores, which are associated with lower rates of remission or response (29); on the contrary, studies from clinical practice use mostly the PMS or the simple clinical colitis activity index. Nevertheless, our results reassure the effectiveness of tofacitinib in UC in clinical practice, even in highly refractory patients, both in the short-term and longterm.

One of the advantages of tofacitinib is its potential for the treatment of other disease-associated EIM and IMID (30). Regarding EIM or IMID active at tofacitinib start, we observed that most of those who are considered to depend on disease activity became inactive under tofacitinib treatment. Our results agree with those previously published from the OCTAVE 1 and 2 trials (31).

Regarding safety, the prevalence and the incidence of adverse events in our clinical practice study was similar to those previously described for tofacitinib, and there was no new safety signal from clinical practice (14,32). In this sense, the only major cardiovascular event occurred in a patient with risk factors; regarding thromboembolic events, 3 of 4 patients had previous risk factors, and to have active UC is also a risk factor by itself.

The American Journal of GASTROENTEROLOGY

Our study has some limitations inherent to its retrospective nature and clinical practice design. There was no predefined protocol for the management of these patients and the recording of information, but it was retrospectively collected from the information registered in the medical records. Endoscopic evaluations at follow-up were lacking in most patients and were only available in those where the evolution after initiation of treatment was not good (as is often the case in clinical practice). Analytical data, such as fecal calprotectin, were not routinely performed but at the discretion of the clinicians responsible for the care of the patients and were therefore not available for all patients. Finally, there could be a selection bias, as a consequence of the retrospective inclusion, of only those patients remembered by the investigator; however, investigators were asked to include all patients who met the criteria at their center, contacting, if necessary, the hospital pharmacy, so we believe that recall bias should not have had a negative impact on the reliability of our results.

However, our study has also several strengths because, to our knowledge, this is the largest study and with the longest follow-up published up to now on the effectiveness of tofacitinib on UC in real life. Accordingly, we could assess the durability of tofacitinib treatment, relapse rate, and response to dose escalation. Furthermore, we were able to describe how tofacitinib is used in clinical practice, its influence on EIM and IMID, and its safety profile in a large cohort in clinical practice.

In conclusion, tofacitinib is relatively effective for UC treatment, even in a highly refractory cohort. A relevant proportion of patients discontinue the drug over time, mainly during the first 6 months and due to primary failure. A relevant proportion of the patients who achieve remission after induction relapse over time, although dose escalation can recapture remission in 60% of them. More than 40% of patients maintain the 10 mg bid dose in clinical practice. Finally, safety was consistent with the known profile of tofacitinib.

CONFLICTS OF INTEREST

Guarantor of the article: María Chaparro, MD, PhD and Javier P. Gisbert, MD, PhD.

Specific author contributions: M.C., D.A., and J.P.G.: study design, data collection, data analysis, data interpretation, and writing the manuscript. Rest of authors: patient inclusion. All authors approved the final version of the manuscript. M.C. and J.P.G. are the guarantors of the article.

Financial support: This is an independent study (ISR) financed by a Pfizer Independent Medical Grant (ID: 66,519,171). Pfizer had no access to clinical data and was not involved in study design, statistical analysis, and manuscript writing.

Potential competing interests: H.A.: speaker for and has received education funding from Janssen, Abbvie, Adacyte, Takeda, Ferring, Nutricia, Falk, and MSD. M.B.-d.A.: speaker, consultant, and advisory member for and has received research funding from MSD, AbbVie, Janssen, Kern Pharma, Celltrion, Takeda, Gillead, Celgene, Pfizer, Sandoz, Biogen, Fresenius, Ferring, Faes Farma, Dr. Falk Pharma, Chiesi, Gebro Pharma, Adacyte, and Vifor Pharma. F.B.: speaker, consultant, and advisory member for and has received research funding from MSD, Abbvie, Takeda, Janssen, Pfizer, Biogen, Amgen, Galapagos, Ferring, Faes Farma, Tillotts Pharma, Chiesi, and Vifor Pharma. M.M.B.-W.: educational activities, research projects, scientific meetings, and advisory boards sponsored by MSD, Ferring, Abbvie, Janssen, Biogen, and Takeda.

M.C.: speaker and consultant for and has receivedresearch or education funding from MSD, Abbvie, Hospira, Pfizer, Takeda, Janssen, Ferring, Shire Pharmaceuticals, Dr. Falk Pharma, Tillotts Pharma, Biogen, Gilead, and Lilly. A.F.-C.: speaker and education funding from Dr. Falk, Janssen, Takeda, and Chiesi y Pfizer. M.J.G.: financial support for travelling and educational activities from Janssen, Pfizer, Abbvie, Takeda, Kern Pharma, Faes Farma, and Ferring. J.P.G.: speaker, consultant, and advisory member for and has received research funding from MSD, Abbvie, Pfizer, Kern Pharma, Biogen, Mylan, Takeda, Janssen, Roche, Sandoz, Celgene/ Bristol Myers, Gilead/Galapagos, Lilly, Ferring, Faes Farma, Shire Pharmaceuticals, Dr. Falk Pharma, Tillotts Pharma, Chiesi, Casen Fleet, Gebro Pharma, Otsuka Pharmaceutical, Norgine, and Vifor Pharma. Á.H.: received grants for conference attendance and conference and consultancy fees from Abbvie, Takeda, Janssen, MSD, Pfizer, Ferring, Falk, and Tillots. R.E.M.: scientific advice, research support, and/or training activities from Pfizer and Janssen y Ferring. N.M.M.: support for attending meetings and speaker fees and consulting fees from Abbvie, Janssen, Takeda, Ferring, Chiesi, Dr. Falk Pharma, and Tillotts Pharma. M.P.: speaker for or has received research funding from Takeda, Abbvie, and Janssen. A.R.-C.: speaker for Takeda. B.S.: scientific advice, research support for and/or training activities from Abbvie, FAES, Chiesi, Dr. Falk, MSD, Tillots Pharma, Khern Pharma, Janssen, Pfizer, and Takeda. M.S.: speaker for and has received research or education funding from Janssen, Abbvie, Pfizer, Takeda, Ferring, and Tillots. R.V.: scientific advice, research support for and/or training activities from AbbVie, Janssen, MSD, Pfizer, FAES-FARMA, Ferring, Shire, and Takeda. The rest of the authors have nothing to declare.

Data availability: The data underlying this article will be shared upon reasonable request to the corresponding author.

Study Highlights

WHAT IS KNOWN

- ✓ Tofacitinib is effective in ulcerative colitis in clinical trials.
- ✓ Treatment survival, effectiveness, and safety in clinical practice remains poorly known.

WHAT IS NEW HERE

- Tofacitinib is relatively effective for ulcerative colitis treatment, even in a highly refractory cohort.
- A relevant proportion of patients discontinue the drug over time, mainly during the first 6 months and due to primary failure.
- A relevant proportion of the patients who achieve remission after induction relapse over time, although dose escalation can recapture remission in 60% of them.
- More than 40% of patients maintain the 10 mg bid dose in clinical practice.
- Safety was consistent with the known profile of tofacitinib.

REFERENCES

- Raine T, Bonovas S, Burisch J, et al. ECCO guidelines on therapeutics in ulcerative colitis: Medical treatment. J Crohns Colitis 2022;16(1):2–17.
- Spinelli A, Bonovas S, Burisch J, et al. ECCO guidelines on therapeutics in ulcerative colitis: Surgical treatment. J Crohns Colitis 2022;16(2):179–89.

The American Journal of GASTROENTEROLOGY

VOLUME 00 | MONTH 2023 www.amjgastro.com

- 3. Chaparro M, Garre A, Ricart E, et al. Short and long-term effectiveness and safety of vedolizumab in inflammatory bowel disease: Results from the ENEIDA registry. Aliment Pharmacol Ther 2018;48(8):839–51.
- Gisbert JP, Chaparro M. Predictors of primary response to biologic treatment [anti-TNF, vedolizumab, and ustekinumab] in patients with inflammatory bowel disease: From basic science to clinical practice. J Crohns Colitis 2020;14(5):694–709.
- Taxonera C, Iglesias E, Muñoz F, et al. Adalimumab maintenance treatment in ulcerative colitis: Outcomes by prior anti-TNF use and efficacy of dose escalation. Dig Dis Sci 2017;62(2):481–90.
- Gisbert JP, Panés J. Loss of response and requirement of infliximab dose intensification in crohn's disease: A review. Am J Gastroenterol 2009; 104(3):760–7.
- Chaparro M, Andreu M, Barreiro-de Acosta M, et al. Effectiveness of infliximab after adalimumab failure in Crohn's disease. World J Gastroenterol 2012;18(37):5219–24.
- Katz L, Gisbert JP, Manoogian B, et al. Doubling the infliximab dose versus halving the infusion intervals in Crohn's disease patients with loss of response. Inflamm Bowel Dis 2012;18(11):2026–33.
- 9. Chaparro M, Panes J, García V, et al. Long-term durability of infliximab treatment in Crohn's disease and efficacy of dose "escalation" in patients losing response. J Clin Gastroenterol 2011;45(2):113–8.
- Chaparro M, Panés J, García V, et al. Long-term durability of response to adalimumab in Crohn's disease. Inflamm Bowel Dis 2012;18(4):685–90.
- Sandborn WJ, Ghosh S, Panes J, et al. Tofacitinib, an oral Janus kinase inhibitor, in active ulcerative colitis. N Engl J Med 2012;367(7):616–24.
- 12. Sandborn WJ, Su C, Sands BE, et al. Tofacitinib as induction and maintenance therapy for ulcerative colitis. New Engl J Med 2017;376(18):1723–36.
- Lucaciu LA, Constantine-Cooke N, Plevris N, et al. Real-world experience with tofacitinib in ulcerative colitis: A systematic review and metaanalysis. Ther Adv Gastroenterol 2021;14:175628482110640.
- 14. Sandborn WJ, Lawendy N, Danese S, et al. Safety and efficacy of tofacitinib for treatment of ulcerative colitis: Final analysis of OCTAVE open, an open-label, long-term extension study with up to 7.0 years of treatment. Aliment Pharmacol Ther 2022;55(4):464–78.
- 15. Panaccione R, Abreu MT, Lazariciu I, et al. Persistence of treatment in patients with ulcerative colitis who responded to tofacitinib therapy: Data from the open-label, long-term extension study, OCTAVE open. Aliment Pharmacol Ther 2022;55(12):1534–44.
- Shimizu H, Fujii T, Hibiya S, et al. Rapid prediction of 1-year efficacy of tofacitinib for treating refractory ulcerative colitis. Intestinal Res 2021; 19(1):115–8.
- Hoffmann P, Globig AM, Thomann AK, et al. Tofacitinib in treatmentrefractory moderate to severe ulcerative colitis: Real-world experience from a retrospective multicenter observational study. J Clin Med 2020;9(7):2177–13.
- Lair-Mehiri L, Stefanescu C, Vaysse T, et al. Real-world evidence of tofacitinib effectiveness and safety in patients with refractory ulcerative colitis. Dig Liver Dis 2020;52(3):268–73.

- Chaparro M, Garre A, Mesonero F, et al. Tofacitinib in ulcerative colitis: Real-world evidence from the ENEIDA registry. J Crohns Colitis 2021; 15(1):35–42.
- Straatmijer T, van Gennep S, Duijvestein M, et al. Real-world clinical and endoscopic outcomes after one year tofacitinib treatment in ulcerative colitis. Eur J Gastroenterol Hepatol 2021;33(10):1288–97.
- Honap S, Chee D, Chapman TP, et al. Real-world effectiveness of tofacitinib for moderate to severe ulcerative colitis: A multicentre UK experience. J Crohns Colitis 2020;14(10):1385–93.
- Weisshof R, Golan MA, Sossenheimer PH, et al. Real world experience with tofacitinib in IBD at a tertiary center. Dig Dis Sci 2019;64(7): 1945–51.
- Biemans VBC, Sleutjes JAM, de Vries AC, et al. Tofacitinib for ulcerative colitis: Results of the prospective Dutch initiative on crohn and colitis (ICC) registry. Aliment Pharmacol Ther 2020;51(9):880–8.
- Deepak P, Alayo QA, Khatiwada A, et al. Safety of tofacitinib in a realworld cohort of patients with ulcerative colitis. Clin Gastroenterol Hepatol 2021;19(8):1592–601.e3.
- Avni-Biron I, Bar-Gil Shitrit A, Koslowsky B, et al. Short-term effectiveness and safety of tofacitinib in ulcerative colitis–real world data from tertiary medical centers in Israel: Tofacitinib in ulcerative colitis. Dig Liver Dis 2022;54(2):192–7.
- 26. Hernández Martínez A, Navajas Hernández P, Martín Rodríguez MdM, et al. Efficacy and safety of tofacitinib in the treatment of ulcerative colitis: Real-life experience in andalusia. Revista Española de Enfermedades Digestivas 2022;114(9):516–21.
- 27. Dubinsky MC, Magro F, Steinwurz F, et al. Association of C-reactive protein and partial Mayo score with response to tofacitinib induction therapy: Results from the ulcerative colitis clinical program. Inflamm Bowel Dis 2022:izac061.
- Sands BE, Armuzzi A, Marshall JK, et al. Efficacy and safety of tofacitinib dose de-escalation and dose escalation for patients with ulcerative colitis: Results from OCTAVE open. Aliment Pharmacol Ther 2020;51(2): 271–80.
- Macaluso FS, Maida M, Ventimiglia M, et al. Factors affecting clinical and endoscopic outcomes of placebo arm in trials of biologics and small molecule drugs in ulcerative colitis: A meta-analysis. Inflamm Bowel Dis 2019;25(6):987–97.
- Wang Y, Wan Z, Jin R, et al. Tofacitinib for extraintestinal manifestations of inflammatory bowel disease: A literature review. Int Immunopharmacology 2022;105:108517.
- Rubin DT, Reinisch W, Greuter T, et al. Extraintestinal manifestations at baseline, and the effect of tofacitinib, in patients with moderate to severe ulcerative colitis. Ther Adv Gastroenterol 2021;14: 175628482110057.
- 32. Rubin DT, Modesto I, Vermeire S, et al. Vermeire|SéverineWorldwide post-marketing safety surveillance experience with tofacitinib in ulcerative colitis. Aliment Pharmacol Ther 2022;55(3):302–10.

© 2023 by The American College of Gastroenterology

The American Journal of GASTROENTEROLOGY